Author's response to reviews

Title: MicroRNA profile in very young women with Breast Cancer

Authors:

Maria Peña-Chilet (mariapch84@gmail.com)
Maria T Martinez (maltemartinez3@yahoo.es)
Jose A Perez-Fidalgo (japfidalgo@msn.com)
Lorena Peiró-Chova (lorena.peiro@uv.es)
Eduardo Tormo (eduardo.tormo@uv.es)
Elisa Alonso-Yuste (elisa.alonso@uv.es)
Beatriz Martinez-Delgado (bmartinezd@isciii.es)
Pilar Eroles (sastago@hotmail.com)
Joan Climent (joan.climent@gmail.com)
Octavio Burgues (burgues_oct@gva.es)
Jaime Ferrer-Lozano (jaime.ferrer@uv.es)
Ana Bosch (anaboschcampos@gmail.com)
Ana Lluch (lluch_ana@gva.es)
Gloria Ribas (gribasdespuig@gmail.com)

Version: 4
Date: 7 July 2014

Author's response to reviews:

MS: 1444141194115678
Research article
MicroRNA profile in very young women with Breast Cancer
Maria Peña-Chilet, Maria T Martinez, Jose A Perez-Fidalgo, Lorena Peiró-Chova, Sara S Oltra, Eduardo Tormo, Elisa Alonso-Yuste, Beatriz Martinez-Delgado, Pilar Eroles, Joan Climent, Octavio Burgues, Jaime Ferrer-Lozano, Ana Bosch, Ana Lluch and Gloria Ribas

BMC Cancer

Dear Dr Melissa Troester:

Thank you for the consideration of our study in BMC Cancer. As editor required, we are sending the three last petitions for a formal acceptance of our manuscript.

-------------------
"Please clarify whether you have consent to publish from participants. If you do not, please anonymize data on page 5. It appears that you have provided individual BMI values and ages which could compromise anonymity of participants."

Thanks to the editor for remarking this inattention. We have anonymized data on page 5, so that individual BMI values are no longer displayed.
"Please include the full name of the institutional ethics committee which approved your study."

We have included the full name of the institutional ethics committee that approved our study on page 4. (The Institutional Health Incliva-Hospital Clinico Ethics Committee).

"Please ensure that your data deposited in the GEO database is released on publication of this manuscript."

Data has been deposited in GEO database, as we explained on page 6, with accession number GSE48088. The data will be released as soon as we receive a formal manuscript acceptance.

------------------

Sincerely,

Dr. Gloria Ribas